Author's response to reviews

Title: Burnout and training satisfaction of medical residents in Greece: will the European Work Time Directive make a difference?

Authors:

Pavlos Msaouel (msaouel@gmail.com)
Nikolaos C Keramaris (nkeram2002@yahoo.gr)
Athanasios Tasoulis (tasoulisathanasios@yahoo.gr)
Dimitrios Kolokythas (dr.dimko@gmail.com)
Nikolaos Syrmos (milanako76@yahoo.gr)
Nikolaos Pararas (npararas@med.uoa.gr)
Eleftherios Thireos (ethireos@hol.gr)
Christos Lionis (lionis@galinos.med.uoc.gr)

Version: 4 Date: 28 June 2010

Author's response to reviews:

Small typographical and language revisions